Author's response to reviews

Title: The caregiving perspective in heart failure: a population based study.

Authors:

Patricia M Davidson (patriciamary.davidson@uts.edu.au)
Amy Abernethy (amy.abernethy@duke.edu)
Phillip J Newton (phillip.newton@uts.edu.au)
Katherine Clark (Katherine.Clark@calvarymater.org.au)
David C Currow (david.currow@flinders.edu.au)

Version: 2 Date: 23 July 2013

Author's response to reviews:

Thank you for the useful and constructive comments of the reviewers. Please find attached detailed comments for the reviewers and amended sections are highlighted in red in the manuscript.